Author's response to reviews

Title: Coronary Artery Disease Evaluation in Rheumatoid Arthritis (CADERA). Quantitative MRI to define mechanisms of cardiovascular co-morbidity in patients with early Rheumatoid Arthritis and to measure the effect of biological therapy: study protocol for a randomized controlled trial.

Authors:

Bara Erhayiem (b.erhayiem@leeds.ac.uk)
Sue Pavitt (s.pavitt@leeds.ac.uk)
Paul D Baxter (p.d.baxter@leeds.ac.uk)
Jacqueline Andrews (j.andrews@leeds.ac.uk)
John P Greenwood (j.greenwood@leeds.ac.uk)
Maya H Buch (m.buch@leeds.ac.uk)
Sven Plein (s.plein@leeds.ac.uk)

Version: 3 Date: 13 October 2014

Author's response to reviews:

Dear Jessie McGavin

Please find attached a cover letter containing the response to reviewers, which accompanies our revised manuscript.

Kind regards

Dr Bara Erhayiem